Author's response to reviews

Title: Alternative and antioxidant therapies used by a sample of infertile males in Jordan: a cross-sectional survey

Authors:

Sanaa K Bardaweel (s.bardaweel@ju.edu.jo)

Version: 4 Date: 9 June 2014

Author's response to reviews: see over
Dear Sir,

Thank you for your letter dated 29th-May-2014, concerning the manuscript entitled “Alternative and antioxidant therapies used by a sample of infertile males in Jordan: a cross-sectional survey”. The author studied your comments carefully and made major corrections, which she hopes would meet with your approval. Please consider my point-by-point responses to the reviewers in the following text.

* Title changed in response to copyediting.
**Editorial Comments:**

We recommend that you ask a native English speaking colleague to help you copyedit the paper. If this is not possible, you may need to use a professional language editing service. For authors who wish to have the language in their manuscript edited by a native-English speaker with scientific expertise, BioMed Central recommends Edanz (www.edanzediting.com/bmc1). BioMed Central has negotiated a 10% discount to the fee charged to BioMed Central authors by Edanz.

**Our response:**

The author would like to thank the respected editor for his valuable time and effort to provide significant suggestions that enhance the manuscript. The author did send the manuscript to a professional language editing service to improve its readability. Edanz provided full copyediting service and a much more readable version of the manuscript. The author would like to thank the editor again for suggesting this professional service that will be of great help to better construct future manuscripts.